Reviewer #1:

**Scientific Quality**: Grade A (Excellent)

**Language Quality**: Grade A (Priority publishing)

**Conclusion**: Accept (High priority)

**Specific Comments to Authors**:

A very interesting description of a patient with GPA, demonstrating how crucial it is to repeat immunological tests in this case, such as ANCA and anti-PR3 antibodies, to avoid misdiagnosis and further consequences, such as the omission of immunosuppressive treatment. A practical description for clinicians.

**Answering Reviewers**:

We extend our sincere gratitude to the reviewers for acknowledging the merit of this paper.